Author's response to reviews

Title: Supracervical hysterectomy versus total abdominal hysterectomy: perceived effects on sexual function

Authors:
Jyot Saini (jsaini@iupui.edu)
Edward Kuczynski (kuczye01@med.nyu.edu)
Herbert F. Gretz III (hfgiii@aol.com)
E. Scott Sills (dr.sills@ivf.com)

Version: 9 Date: 8 Jan 2002

The co-authors have reviewed the 31-point checklist and attest to the fact that the work, in its present form, complies fully. As we indicate in the paper, none of us has any competing interest to declare with this research or its conclusions.
For and on behalf of the authors,
E. Scott Sills, MD